Math 132 Winter 2004 R. Schultz

UPDATED GENERAL INFORMATION — MARCH 9, 2004

Here are the results for the second examination (P = total points possible):

GRADE CUTOFF POINTS

P: 100 A: 84 B: 62 C: 42 D: 26

The median score was 77, and the average was approximately 70. Requests for reconsideration must be submitted in writing no later than the date and time of the final examination. Any explanations of reasons should be brief and deal with the specific mathematical points involved. Solutions for the examination problems have been posted in the file exam2a.pdf in the course directory. Solutions for the exercises in the final assignment have been posted in the file solutions5.pdf in the course directory.

The final examination will consist of ten questions that are similar in nature to the sorts of questions on the first and second examinations. These problems will be about 70 per cent computational and 30 per cent theoretical.